Author's response to reviews

Title: Isolated rural general practice as the focus for teaching core clinical rotations to pre-registration medical students.

Authors:

Stephen A Margolis (s.margolis@uq.edu.au)
Llewellyn M Davies (llew.davies@uq.edu.au)
Valmae A Ypinazar (v.ypinazar@uq.edu.au)

Version: 4 Date: 26 June 2005

Author's response to reviews:

Dear Editor

Changes made as requested.

Yours sincerely

Stephen Margolis